Author's response to reviews

Title: Prevalence of Malaria and Anaemia among HIV infected pregnant women receiving Co-trimoxazole prophylaxis - A cross sectional study in One District in Tanzania.

Authors:

Vicent Manyanga (vcenetz@yahoo.com)
Omary M Minzi (Minziobejayesu@gmail.com)
Billy Ngasala (bngasala@muhas.ac.tz)

Version: 2 Date: 2 December 2013

Author's response to reviews:

Dear Editor, We have inserted the conclusion and slightly revised the abstract. We also inserted a conclusion after the discussion section.

With kind regards.

Minzi OMS
Corr Author